Author's response to reviews

Title: A Bleeding Kiss: Intramural haematoma secondary to balloon angioplasty

Authors:

Cheerag Shirodaria (cshirodaria@doctors.org.uk)
William J van Gaal (william@vangaal.org)
Adrian P Banning (adrian.banning@orh.nhs.uk)

Version: 4 Date: 5 June 2007

Author's response to reviews:

The requested changes to the manuscript have been made. Please let me know if any further changes need to be made.

Regards,
Cheerag Shirodaria